

Contents lists available at ScienceDirect

# Sustainable Production and Consumption



journal homepage: www.elsevier.com/locate/spc

# Research article

# Evaluating circular economy strategies for raw material recovery from end-of-life lithium-ion batteries: A system dynamics model

# Bhanu Pratap<sup>a</sup>, T.V. Krishna Mohan<sup>b,\*</sup>, R.K. Amit<sup>a</sup>, Shankar Venugopal<sup>c</sup>

<sup>a</sup> Decision Engineering & Pricing (DEEP) Lab, Department of Management Studies, Indian Institute of Technology Madras, Chennai, 600036, India

<sup>b</sup> University of Exeter Business School, Exeter, EX4 4PU, UK

<sup>c</sup> Mahindra Research Valley, Mahindra & Mahindra, Chennai, 603 004, India

# A R T I C L E I N F O

Keywords: EOL LIB management Recycling Repurposing System dynamics

# ABSTRACT

Across the globe, with the increasing emphasis on decarbonization, lithium-ion battery (LIB) demand for mobility (which serves as a power source for electric vehicles) and stationary energy storage sector (SESs) increases, which generates a large stock of end-of-life (EOL) LIBs. Continually increasing the stock of EOL LIB having different LIB variants necessitates the development of efficient circular economy (CE) strategies (recycling and repurposing) to recover raw materials contained in them. Focusing on different CE strategies, we develop a system dynamics model to address the complexity of the raw material recovery process by analyzing the interrelationship between collection rate (government), EOL LIB variant mix (consumer preference), and EOL LIB allocation to recycling and repurposing (Battery OEMs). Our analysis reveals that a high EOL collection rate and recycling reduces the raw material (Lithium (Li), Nickel (Ni), and Cobalt (Co)) demand by 2%-17% based on LIB variant proportion in EOL LIB stock. We observe thrice higher Co recovery and 1.5 times higher Ni recovery in material-rich battery chemistries as compared to others. Repurposing delays the raw material recovery but reduces LIB's demand for SESs. In addition, we observe that the repurposed EOL LIB supply increases the recyclable EOL LIB supply by 0.027-0.2 million units at the end of 2030. Hence, it is imperative for emerging economic countries like India, with scarce strategic raw materials sources and increasing demand for LIB from mobility and SES sectors, to frame policies that incentivize the collection and EOL handling process infrastructure and prioritize between recycling and repurposing of EOL LIBs.

# 1. Introduction

Mobility transitions have become the focus of practitioners and scholars with a growing emphasis on the decarbonization of transportation systems (Abergel et al., 2020; Bibra et al., 2022). The proliferation of electric vehicles (EVs) in the transportation sector is imperative to induce a paradigm shift in the mobility sector, as EVs are considered zero-tailpipe emission vehicles. Batteries power EVs, and among all commercially available batteries, LIBs are most suitable to power EVs based on their superior performance characteristics (Nitta et al., 2015; Schmuch et al., 2018). Global LIB demand is forecasted to grow by 30%–35% to reach 4.7 terawatt-hours (TWh) in 2030 (Fleischmann et al., 2023). LIB demand from the automotive sector and stationary energy storage (SES) contribute 70%–80% of global LIB demand.

Growing demand for LIBs generates a large pool of end-of-life (EOL) LIB when LIB nominal capacity reaches 80% of its original capacity (Goodenough, 2012; Morse, 2021). LIB life varies from

3–10 years based on driving conditions, LIB variants, charging frequency, and warranty offered by battery original equipment manufacturers (OEMs) (Richa et al., 2014; Zubi et al., 2018; Beuse et al., 2020). NITI Aayog, a strategy think tank in India, forecasted that total cumulative LIB demand from the electric mobility sector will reach 381 GWh in 2030 which will generate 70–80 GWh EOL LIB in 2030. Abergel et al. (2020) forecast that 100–120 GWh of LIBs will reach EOL in 2030 based on the difference in scenarios taken by the International Energy Agency (IEA).

EOL LIBs are collected from the consumers and transported to different locations for further processing. EOL LIBs are transported through various means, like trucks, ships, etc., to other sites based on processing facility location, ease of processing, and favorable government policies (Schulz-Mönninghoff and Evans, 2023; Lander et al., 2021; Slattery et al., 2021). The collection rate depends on the efficiency of

\* Corresponding author.

https://doi.org/10.1016/j.spc.2024.07.027

Received 29 February 2024; Received in revised form 10 July 2024; Accepted 29 July 2024 Available online 7 August 2024

*E-mail addresses:* bhanupratap@alumni.iitm.ac.in (B. Pratap), k.m.tv@exeter.ac.uk (T.V.K. Mohan), rkamit@iitm.ac.in (R.K. Amit), venugopal.shankar@mahindra.com (S. Venugopal).

<sup>2352-5509/© 2024</sup> The Author(s). Published by Elsevier Ltd on behalf of Institution of Chemical Engineers. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

collection processes (Vlachos et al., 2007), mode of transportation, and government regulations (Jacoby, 2019; Dunn et al., 2022).

Collected LIB undergoes stringent quality checks based on the state of health (SOH), state of charge (SOC), etc., of LIB before allocation to different circular EOL management processes like remanufacturing, repurposing, and recycling (Chirumalla et al., 2023). Remanufactured EOL LIB undergoes stringent quality checks before being reused in EVs (Albertsen et al., 2021; Chirumalla et al., 2023; Schulz-Mönninghoff and Evans, 2023). Due to lack of standard quality check methods and regulations, we only consider the recycling and repurposing process for managing EOL LIB. EOL LIB can be either recycled to obtain critical raw materials like lithium (Li), cobalt (Co), and nickel (Ni) (P. Ryan; The Economist, May 13, 2021), or it can be repurposed as second-use LIB for other applications (D. Holger et al.; The Wall Street Journal, June 13, 2022) like stationary storage systems, frequency regulation, renewable integration, etc., (Chen et al., 2019; Liu et al., 2020; White et al., 2020; Dunn et al., 2022).

Raw material recovery from EOL LIB through recycling depends on the recycling process efficiency (Dunn et al., 2022; Liu et al., 2023), battery mix in total LIB demand (Jiang et al., 2021; Kamath et al., 2023), LIB capacity (Shafique et al., 2023), and recycling capacity (Georgiadis and Athanasiou, 2013), whereas the quality of collected EOL LIB, maturity of the repurposing process, demand for repurposed LIB, and repurposing capacity impact the LIB second-use (Vlachos et al., 2007; Schulz-Mönninghoff et al., 2021; Dunn et al., 2022; Alšauskas et al., 2024). Repurposing enables EOL LIB's use in cascade application (Bobba et al., 2019) and improves the LIB life but, at the same time, it delays the recycling process which soars the demand for the virgin raw material.

Along with this, varying collection rates due to the nascent stage of growth of the collection infrastructure (Georgiadis and Besiou, 2010; Rizova et al., 2020), LIB variant mix in continually increasing EOL LIB stock (Dunn et al., 2022), lack of government policies (Alsauskas et al., 2023; Alšauskas et al., 2024), and EOL LIB handling infrastructure (Georgiadis and Athanasiou, 2013) bring complexity in planning and managing different EOL recovery strategies. The factors mentioned above interact with each other through a nonlinear multiloop feedback system, and the delay in realization of the impact of interacting among variables on raw material recovery adds further complexity in the allocation of EOL LIB to different EOL LIB handling process. Therefore, in-depth analysis is required to explore the tradeoff between EOL LIB handling processes impacted by EOL LIB collection rate and infrastructure, battery mix of EOL LIB, and EOL LIB handling infrastructure. Hence, we develop a system dynamics (SD) model to understand the complexity of recovery of raw material from EOL LIB by conducting a brief analysis among the following directions:

- How does variation in EOL LIB collection rate impact the raw material recovery from EOL LIB?
- How will the consumer preferences for LIB variants impact the recovery of strategic raw material from EOL LIB?
- How does the priority in allocating collected EOL LIB for LIB recycling and repurposing impact the raw material recovery market?

The system dynamics methodology is widely used for understanding the complexity of raw material recovery from EOL vehicles and LIBs by using different circular management strategies (Georgiadis and Besiou, 2010; Georgiadis and Athanasiou, 2013; Bobba et al., 2018b; Alamerew and Brissaud, 2020; Mohan and Amit, 2021; Weigl and Young, 2023; Kamath et al., 2023). To the best of our knowledge, this is the first attempt to apply the system dynamics model to analyze the trade-off between circular economy strategies' impact on raw materials recovery from EOL LIB with varying collection rates and different LIB proportions in EOL LIB stock. We consider the impact of varying LIB variant proportions in EOL LIB stock, minimum collection ratio, and impact of tradeoff of allocation different ratio of EOL LIB to recycling and repurposing on raw material recovery and availability of repurposed LIB. The model results highlight that raw materials recovery from EOL LIBs increased by twice when the collection rate of EOL LIBs increased from 50% to 90%. Increasing the proportion of lithium iron phosphate (LFP) batteries in collected EOL LIB reduces the recovery of strategic raw materials Li, Ni, and Co from the EOL LIB as LFP does not contain Ni and Co. Co recovery is three times, and Ni recovery is around 1.5 times higher in high cobalt scenarios compared to LFP scenarios. Apart from this, improving the repurposing of EOL LIBs minimizes the demand for new LIBs for stationary energy storage but reduces the raw material recovery, as repurposing extends the LIB life. Repurposed LIB variant availability depends on the battery mix of EOL LIB, repurposing infrastructure, and EOL LIB collection rate. Recyclable LIB supply increased from EOL repurposed LIB supply (depends on repurposed LIB life in different appliances), which improves the raw material recovery that eases the pressure on exploring new raw material mines to fulfill the increasing demand for the booming electric mobility market

This paper is organized as follows: Section 2 reviews the literature on the circular economy strategy used for managing EOL LIB and raw material recovery. Section 3 details the system dynamic model. Section 4 discusses the simulation results, and Section 5 provides a discussion of the results. Section 6 concludes the paper.

# 2. Literature review

EOL LIB can be recycled to obtain valuable raw materials or be used as a second life LIB as EOL LIB landfill pollutes the environment and is hazardous to human beings. This section focuses on literature related to different circular economy strategies for managing EOL LIB. We also discuss the literature on the usefulness of system dynamics in managing the complex EOL LIB recovery management system that helps in raw material recovery.

#### 2.1. Circular economy strategies for LIB

According to the waste management hierarchy, repair, remanufacturing, repurposing, and recycling are widely used circular economy (CE) strategies for handling EOL LIB. In remanufacturing and repurposing, cells or modules that lose 80% of their nominal capacity are replaced with new cells or modules in the battery pack (repurposing needs change in battery management system based on applications) and then used other applications (Chen et al., 2019; Schulz-Mönninghoff and Evans, 2023). Remanufactured LIB undergoes strict quality checks before use in EV. Recycling helps extract raw materials from EOL LIB, reducing the demand for virgin raw materials for LIB. A detailed explanation of repurposing, recycling, and the impact of tradeoffs between recycling and repurposing raw material recovery is explained below.

#### 2.1.1. LIB recycling

Physical materials separation, pyrometallurgical recovery, hydrometallurgical, direct recycling, and biological metals reclamation are commonly used for LIBs recycling (Chen et al., 2019; Harper et al., 2019; Chen et al., 2022; Makwarimba et al., 2022). In Pyrometallurgical recycling, LIBs are fed to a high-temperature furnace where the burning of electrolytes and plastic in the batteries provides heat; graphite/carbon and aluminum are oxidized; Co, Ni, copper (Cu), and iron is recovered as matte, and the rest of the materials end as slag. Matte is further processed through acid leaching followed by solvent extraction to recover Ni and Co (Chen et al., 2019; Sommerville et al., 2021). In Hydrometallurgical recycling, EOL LIBs are shredded, and a lowtemperature calcination process is used to burn binder and electrolyte. After that leaching process, solvent extraction is performed to obtain lithium carbonate, Co, and Ni compounds (Sommerville et al., 2021; Makwarimba et al., 2022). In the direct recycling process, EOL LIBs are shredded, and then several physical separation processes are applied to

recover the cathode, anode, plastics, and metals. The recovered cathode then undergoes lithium replenishment in the cathode to produce a new cathode used for EVs (Chen et al., 2019; Harper et al., 2019). The hydrometallurgical process recovers Li, Co, and Ni, whereas only Co and Ni are recovered from Pyrometallurgical recovery (Gaines, 2014; Dai et al., 2019).

LIBs are recycled to obtain critical raw materials embedded in the electrodes and electrolytes when they reach EOL (Harper et al., 2019; Kamran et al., 2021). Hence, it reduces the EOL LIB landfills and eases pressure on new raw materials mines needed to cater to rising LIB demand (Chen et al., 2019). Shafique et al. (2022) use multidimensional scenario analysis to predict the quantity of recovered Li, Ni, and Co in China and the U.S.. Authors state that in 2030, 5–7 kilotons (kt) Li, 35–60 kt Ni, and 4–6 kt Co are recovered from EOL LIB in China, whereas in the U.S., 2.3–2.6 kt Li, 16–26 kt Ni, and 3–4 kt Co is recovered. Liu et al. (2023) explores that recycling of EOL LIB will cover 27.4–42.0% Li, 50.4–77.5% Co, and 50.4–77.6% Ni demand for new LIBs. Zeng et al. (2022) develop different scenarios for analyzing the impact of secondary raw materials on primary raw material demand. The authors forecast that total secondary Co stock would reach 3680 kt in 2020–2050.

#### 2.1.2. Repurposing of LIBs

EOL LIBs are repurposed after stringent quality checks. EOL LIBs are dismantled at the module or cell levels to identify defective parts. Damaged LIB parts are replaced after inspection, and a new battery management system is included in the LIB pack (based on the application) to develop repurposed LIBs (Bobba et al., 2019; Chen et al., 2019; Alfaro-Algaba and Ramirez, 2020). Repurposed EOL LIB can be used as stationary energy storage, electricity grid frequency regulation service, behind-the-meter (BTM) energy storage, load leveling, and many more applications (Heymans et al., 2014; White et al., 2020; Khowaja et al., 2022). Bloomberg New Energy Finance has estimated the potential of the global second-life batteries market is expected to be about 26 gigawatt hours (GWh) in 2025 (equivalent to about 47% of the global LIB supply in 2015). Engel et al. (2019) predict that second-life supply will vary from 117-227 GWh/year based on different scenarios. Xu et al. (2023) estimates that the cumulative capacity of the second-use battery will reach 14.8-31.5 terawatt-hours (TWh) in 2050 when the second-use battery is utilized for ten years. 4R Energy Corp repurposed EOL Nissan Leaf batteries for stationary energy storage.

Although a high volume of EOL LIBs will be available for seconduse by 2050, complex and customized battery design, decreasing new LIB prices, and customer acceptance for repurposed LIBs limits the EOL LIB use as second-use LIB (Neubauer et al., 2015; Martinez-Laserna et al., 2018; Engel et al., 2019; Zhu et al., 2021). Martinez-Laserna et al. (2018) conducts a comprehensive literature review on LIB's secondary use. Engel et al. (2019) analyze factors impacting the EOL LIB repurposing. Authors state that customization of LIB based on EV models decreases cost parity among new and repurposed LIB. However, the lack of regulatory policy on LIB reuse and inadequate repurposed LIB quality assessment standards limit the EOL LIB repurposing.

# 2.1.3. LIB repurposing-recycling trade-offs

EOL LIB can be recycled to obtain critical raw materials like Li, Co, and Ni, or it can be second use for other applications like stationary storage systems, frequency regulation, renewable integration, etc., (Chen et al., 2019). Hence, there is always a trade-off between LIB recycling and repurposing (Krishna Mohan et al., 2021). Repurposing extends the LIB life to 5–10 years based on usage but delays the EOL LIB recycling, reducing the recycled material supply (Bobba et al., 2018a). Along with delaying the recycled material supply, the recycling-repurposing trade-off has different economic impacts. Li et al. (2018) develop a Mixed integer nonlinear programming (MINLP) model to analyze the impact of integrating remanufacturing and recycling infrastructure to a standalone LIB manufacturing facility on supply chain profit. The authors state that including remanufacturing will enhance supply chain profit by 30.93% for a standalone manufacturing facility and 9.81% for an integrated manufacturing-recycling facility. Jiang et al. (2021) develop a Material flow analysis model to analyze the economic impact of cascade utilization and material recovery (through recycling only). The authors find that cascade utilization of EOL LIB creates a monetary value of 147.8 billion US dollars (USD) by 2040, twice the value (76.9 billion USD) created through only material recovery. Reinhardt et al. (2019) suggests that the high costs issue with LIBs can be partly resolved through the LIBs' second use that enables the development of less-energy-intensive storage systems, prolonging the battery life and closing the resource loop.

# 2.2. System dynamics modeling

Multiple stakeholders are involved in EOL LIB recovery strategies. They are involved in each stage of the recovery process, from the collection of EOL LIB to recycling EOL LIB to get recycled products. All the stages are interlinked, and different factors impacting different stages are interconnected, bringing complexity to different recovery processes. The system dynamics modeling methodology developed by Prof J. W Forrester is a widely used methodology to analyze the complex system (Abbas and Bell, 1994; Berends and Romme, 2001; Georgiadis and Besiou, 2008; Feng et al., 2013; Wang et al., 2014; Nieuwenhuijsen et al., 2018; Pratap et al., 2019; Bhanu et al., 2024; Pratap et al., 2023). Wang et al. (2014) develop a system dynamics model to analyze the impact of passenger car life span on EOL vehicle waste in Belgium. The authors find that the ELV reuse and recovery rate increases with a decreasing passenger car life span. Li et al. (2023) build a system dynamics model to analyze the impact of subsidy on EV sales. Authors state that under static subsidy scenarios, acquisition subsidy has a higher influence on EV sales than R&D subsidy, but under fixed subsidy budget, R&D subsidy has a more significant impact on EV sales than acquisition subsidy. Li et al. (2020) uses a gamebased system dynamics model to analyze the effect of a deposit-refund scheme on EV recycling. The authors identify that the collection rate of used EVs increases with an increase in the deposit-refund scheme but is constrained by government subsidy budget allocation. Model results highlight that consumers' environmental awareness increases the collection rate of used EVs.

#### 2.3. Overview of relevant literature

In summary, many studies present different methods of LIB recycling, their impacts on raw material recovery, and the economic impact of LIB recycling. Similarly, studies related to repurposing explain the impact of repurposing on the availability of repurposed LIB for secondary use and the financial benefits of repurposing. Table 1 provides an overview of the literature relevant to our paper. Table 1 highlights that only a few studies focus on exploring the tradeoff between recycling and repurposing on raw material recovery and availability of second-use LIB, EOL repurposed LIBs impact on raw material recovery, the effect of delay on collection, recycling, and repurposing infrastructure on raw material recovery, and impact of charging frequency on LIB replacement. Along with the dilemma of recycling and repurposing and the lack of infrastructure, different LIB variant adoption by consumers and raw material composition in LIB variants impact raw material recovery. This study takes a holistic approach backed by SD modeling to explore the impact of the tradeoff between recycling and repurposing on variability in raw material recovery for EOL LIB. This study also examines the time delay impact on infrastructure development, variable collection rate, varying LIB variants market share, and raw material composition in LIB variants impact on raw material recovery.

#### Sustainable Production and Consumption 50 (2024) 191-204

# Table 1

Overview of the most rele	evant interature.				
Author	Method	Battery type	Materials accessed	Circular economy strategy	
Bobba et al. (2019)	Material flow analysis	NMC, NCA,	Li, Co	Remanufacturing, Repurposing, Recycling	
Wasesa et al. (2022)	Hybrid simulation (Agent-based and System dynamics	NMC, NCA, LFP	Li, Ni, Co, Cu, Al	Recycling	
Dunn et al. (2022)	Material flow analysis.	NCX (NMC 632, NMC 811, and NCA), LFP	Li, Co, Ni	Recycling	
Kastanaki and Giannis (2023)	Material and substance flow analysis	NCA, LMO, LFP, NMC 111, NMC 532, NMC 622, NMC 811, NMC 955	Li, Co, Ni, Cu	Remanufacturing, Reuse, Recycling	
Shafique et al. (2023)	System dynamics model	NMC, NCA, LFP	Li, Ni, Co, Cu	Recycling	
Liu et al. (2023)	Stock driven model	LCO, NMC, NCA, LMO	Li, Co, Ni, Mn, Cu, Al, Fe	Recycling	
Schulz-Mönninghoff and Evans (2023)	Inductive research approach	NMC	Li, Ni, Mn, Co	Repurposing	
Chirumalla et al. (2023)	Multi-faceted approach	Lithium-ion battery	Focused on second life use of LIB	Remanufacturing, Repurposing, Reuse	

# 3. Methods

In this section, we provide the details of the system dynamics model used in our study. Section 3.1 explains the market setting and provides theoretical support for the system dynamics model developed. Section 3.2 gives the scope of the problem, describes the various relationships identified in the system, provides the system dynamics model, and explains the various decision rules that govern the model.

# 3.1. Market setting

Battery manufacturers are always in a dilemma in EOL LIB allocation for recycling or repurposing when LIB reaches EOL. EOL LIBs allocation to different recovery processes depends on variables like battery collection rate, battery quality, recycling process efficiency, repurposing efficiency, minimum collection quantity, etc. Interconnection and dependency among the abovementioned variables complicate the LIBs recovery process. In our model, government regulations, EOL LIB collection capacity, and recovery process profitability impact the quantity allotted to the recycling process and second use. Collected EOL undergoes a decision process that decides the quantity allotment for recycling and second use. The leftover EOL LIB is sent to the landfill. The quantity allotment to recycling and second use affects the raw material recovery from EOL LIB, process profitability, demand for virgin raw material, and EOL LIB ending as a landfill.

#### 3.2. System dynamics modeling

System dynamics models are a widely used methodology for exploring a complex system involving strategic recycling and repurposing decisions (Georgiadis and Besiou, 2010; Alamerew and Brissaud, 2020; Kamath et al., 2023; Li et al., 2023). Hence, we develop a system dynamics (SD) model to study the trade-offs between EOL LIB recycling and second-use and its impact on raw material recovery and the LIB market. SD modeling consists of various steps, which consist of defining model scope, identifying the causal relationship among variables linked through a multi-loop feedback system, developing of stock and flow (simulation model diagram, and testing the model accuracy and validity.

Table 2

venonin o	absempts desemption.		
S. No.	Name	Description	Elements
1	Battery	Lithium-ion battery Variants	NMC622; NMC811; NCA; LFP
2	Battery Mix	Battery proportion under consideration	NMC622 ratio (BR1); NMC811 ratio (BR2); NCA ratio (BR3); LFP ratio (BR4)
3	Raw material	Chemical composition of electrodes	Lithium (Li); Nickel (Ni); Cobalt (Co)
4	Recycling efficiency	Raw material recycling efficiency	Lithium (RE1); Nickel (Ni); Cobalt (RE2)
5	Raw material price	Price of raw material used in LIB electrodes	Lithium (Lip); Nickel (Nip); Cobalt (Cop)

## 3.2.1. Scope of the model

In this section, we mainly focus on defining model boundaries. As the model explores the impact of the tradeoff between the allocation of EOL LIB to recycling and repurposing, which are predominant in emerging economies, we use data from an emerging economy, i.e., India. We consider EVs of a homogeneous nature (battery electric cars (BEC)) and stationary energy storage (SES) for energy storage for household applications demand for the Indian market. We consider BES and SES devices as they contribute around 80% of total LIB demand (Alsauskas et al., 2023; Alšauskas et al., 2024). We use the subscript option in Vensim<sup>®</sup> Pro Software (Version 6.4E) to assign values to different raw materials and batteries. All the values are calculated every month. Battery ratio is the same for EV sales, SES, collected EOL LIB, recycled LIB, and repurposed LIB. We assume that the demand for LIBs is equivalent to the sales of BECs and SESs. Different LIB variants' demand is calculated by multiplying initial BEC and SES sales by the battery ratio. We use subscripts for variables that enable repetitions of variables without cluttering the model, as subscripts do not appear in sketches (Harrison et al., 2016). Five subscript ranges are represented in the model shown in Table 2. We have summarized the parameter values used to develop the different scenarios in Table 3.

#### 3.2.2. Causal loop diagram

The causal loop diagram (CLD) provides a macro-level representation of the interactions in the system. A causal loop diagram (CLD) acts

Table 3				
Scenarios used in the System dynamics model.				
Model input	Scenarios			
Cathode chemistry sales	Three scenarios taken from International Energy Agency (IEA) EV outlook report defined in Table 4 (Source: Alsauskas et al. (2023), Alšauskas et al. (2024))			
Percentage repurposed	Varies from 10%–90% (Model assumption)			
Percentage recycled	Varies from 10%-90% (Model assumption)			
Collection rate	Varies from 10%–90% (Model assumption)			
Raw material composition	Defined in Table 5 (Source: Olivetti et al. (2017))			



Fig. 1. System dynamics model: Causal loop diagram.

as a visual tool that assists in developing the pictorial representation of a dynamically complex system containing multi-loop, feedbacks, delays, and non-linearity (Senge and Sterman, 1992; Ford, 1999; Vlachos et al., 2007; Sterman, 2010). Model complexity arises due to interconnection and interrelationships among variables that explain model behavior. Model variables are interlinked by a causal linkage and act as feedback in the system (Wolstenholme and Coyle, 1983; Berends and Romme, 2001; Kunc and Morecroft, 2007; Ford, 2018). CLD has three main components: causal links (shown by arrows) that explain the relationship among variables, and loop polarity represented by positive (+) or negative (-) signs that help identify loop polarity. It indicates the change in the dependent variable with respect to the independent variable when all other variables are kept constant, and the loop identifier helps identify the nature of the loop – balancing or reinforcement.

In this section, we explore the causal relationship among variables using a CLD that provides the theoretical background for developing the system dynamics model. Minimum collection rates defined by the government, collection capacity, recycling and repurposing capacity, and battery mix are variables interlinked to create a dynamically complex system containing feedback and delays impacting raw material demand.

Fig. 1 shows the causal loop diagram (CLD) to understand the impact of different EOL LIB recovery strategies on raw material demand. CLD represents the causal relationships between various variables in the system. We identify six feedback loops in the system: balancing loops – "EOL LIB collection", "Repurposing profit" and reinforcement loop – "EOL LIB supply", "Repurposing cost", "Recycling cost", "Recycling cost".

• Balancing loop, B1 - EOL LIB collection: Collected EOL battery  $\uparrow \Rightarrow$  Collection capacity increment  $\downarrow \Rightarrow$  Collection capacity  $\downarrow \Rightarrow$ Collected EOL battery  $\downarrow$ 

EOL LIBs are collected through different modes of transportation before processing through different EOL management practices- recycling and repurposing (Lander et al., 2021; Schulz-Mönninghoff and Evans, 2023). The collection of EOL LIB depends on collection capacity (Vlachos et al., 2007). Enhancement of collection capacity leads to an increment in the collected quantity that reduces the landfill, which reduces the harmful impact of dumping the EOL LIB in open areas (Georgiadis and Athanasiou, 2013).

Balancing loop, B2 - Repurposing profit: Repurposed LIB supply
 ↑ ⇒ Repurposed LIB price ↓ ⇒ Repurposing revenue ↓ ⇒ Repurposed profit ↓ ⇒ Repurposed LIB ↓ ⇒ Repurposed LIB supply
 ↓

An increase in collection capacity enhances the LIB repurposing (depending on the quality of the collected EOL LIB), which increases the repurposed LIB supply (Georgiadis and Athanasiou, 2013; Chirumalla et al., 2023). As the supply of repurposed LIB increases, the price of repurposed LIB decreases (due to scaling effects). The reduction in repurposing LIB price reduces the repurposing revenue, which reduces profitability, leading to further capacity contraction and, consequently, less LIB repurposing. The supply of repurposed LIB falls alongside the reduction in repurposing activities.

Reinforcement loop, R1 - EOL LIB supply: EOL LIB ↑ ⇒ Collected EOL LIB ↑ ⇒ LIB repurposing ↑ ⇒ Repurposed LIB ↑ ⇒ Repurposed LIB supply ↑ ⇒ EOL LIB ↑

EOL LIB collection increases with the increasing EOL LIB availability to reduce the EOL LIB landfill (Chen et al., 2019). Collected EOL LIBs are used as repurposed LIBs after the stringent quality check of collected LIBs. Increasing repurposing (utilization of repurposed capacity) enhances the repurposed LIB supply. Repurposed LIB reaches the EOL stage after some time delay (based on application) and grows the EOL LIB availability (Schulz-Mönninghoff and Evans, 2023).

- Reinforcement loop, R2 Repurposing cost: Repurposed LIB ↑
   ⇒ Repurposing cost ↓ ⇒ Repurposed profit↑ → Repurposed LIB ↑
- Reinforcement loop, R3 Recycling cost: LIB recycling ↑ ⇒ Recycling cost ↓ ⇒ Recycling profit ↑ ⇒ LIB recycling ↑
   An increment in the availability of EOL LIB for repurposing reduces the repurposing cost. Cost reduction enhances the profitability of repurposing, resulting in more EOL LIB repurposing. Similarly, recycling costs are reduced with increased LIB recycling, leading to more profitability in the recycling process. As a result of which, more LIBs are recycled (Bobba et al., 2019; Dunn et al., 2022).
- **Reinforcement loop, R4** *Recycling profit*: LIB recycling  $\uparrow \Rightarrow$  Recycled raw material supply  $\uparrow \Rightarrow$  Recycling revenue  $\uparrow \Rightarrow$  Recycling profit  $\uparrow \Rightarrow$  LIB recycling  $\uparrow$

LIB recycling enhances the supply of recycled raw material based on the recycling process efficiency and recycled LIB quantity (Kamath et al., 2023). Growth in the supply of recycled material increases recycling profitability by enhancing the revenue obtained from recycled material. Profitability enhancement brings more recyclers into the process, which results in growth in the recycling process.

# 3.2.3. Stock and flow diagram

The stock and flow diagram (SFD) helps establish mathematical relationships among model variables that assist in understanding model dynamics arising from interaction among variables defined in CLD. Fig. 2 represents the SFD that helps understand the trade-offs between recycling and repurposing. Model variables are categorized into stock, flow, and auxiliary variables. A rectangular box represents stock/state variables changed using flow variables denoted by the valve symbol. An auxiliary variable consists of stock functions, constants, or exogenous inputs. A stock variable represents a level or inventory value at a given period. The flow variables (inflow and outflow) are the rates at which stock variables change and are represented as a function of time. Clouds outside the model boundary are a source or sink of flow variables. It is assumed to have an infinite capacity and supports flow variables.

Table 4

Proportion of LIB variants in total LIB demand.

Jource.	onvetti et al. (20	17, whatset et al. (2023).		
S. No.	Battery variant	High Cobalt Scenario (Values in %)	High Nickel Scenario (Values in %)	LFP
1	NMC622	0.76	0.1	0
2	NMC811	0.13	0.8	0.6
3	NCA	0.11	0.1	0.1
4	LFP	0	0	0.3

Table 5

Element Requirements (Li, Co, Ni) for LIB variants (Units: kg/kWh).

S. No.	Battery Chemistry	Raw material composition		
		Li	Ni	Со
1	NMC622	0.126	0.641	0.214
2	NMC811	0.111	0.75	0.094
3	NCA	0.112	0.759	0.143
4	LFP	0.084	0	0

#### 3.3. Model variables

We classify the model variables in themes such as "EOL LIB supply" (EOL LIB calculation), "EOL LIB collection" (collection capacity adjustments), "EOL LIB recycling" (recycling costs and profit calculation), "EOL LIB repurposing" (repurposing costs and profit calculation), "Learning effect" (LIB production cost variation), and "Raw material" (raw material demand and supply). Different LIB chemistry variants such as NMC 622, NMC 811, NCA, and LFP are considered, and their proportionate demand is represented in Table 4. The change in raw material composition with changing battery variants is described in Table 5. All model variables are explained in detail in the paper's supplementary information.

# 4. Results

The model is simulated for 180 months with INITIAL TIME = 0, FINAL TIME = 180, and TIMESTEP being 0.5 months results are no longer sensitive to the choice of TIMESTEP when we reduce TIMESTEP below 0.5 months). This section explores the impact of variations in LIB variants composition in EOL LIB, collection rate, and EOL LIB allocation to recycling and repurposing. In each defined scenario, we consider the initial recycling and repurposing capacity to be 100 units and the LIB capacity to be 60 kilowatts (kWh). In Scenarios 2, 3, and 4, we considered the EOL LIB collection rate as 50%. In Scenarios 1 and 2, the proportion of EOL LIB allocated to recycling is 80%. In Scenario 1, the collection capacity of EOL LIB varied between 10%-90% (based on Kamath et al. (2020), Wasesa et al. (2022)). In contrast, in Scenario 2, LIB variant proportion in EOL LIB varies based on scenarios given in Table 4. In Scenario 3 and 4, we varied the proportion of EOL LIB allocated to recycling and repurposing (10%-90%). Scenario 1 and 3-4 consider the Low Nickel Scenario of Table 4 for simulation.

# 4.1. Model validation

We follow multiple validation methods provided in the literature to verify the suitability and appropriateness of our model (Barlas, 1989, 1996; Thies et al., 2016). Along with this, we perform structural validity tests to gain confidence in the usefulness, applicability, and simplicity of our model, as there is currently no method available to establish the correctness of the system dynamics model (Ford and Flynn, 2005; Taylor et al., 2010; Sverdrup and Ragnarsdottir, 2016; Wasesa et al., 2022).

We perform structural validity tests to affirm the similarity of our model to a real-world system. We are confident in the structural



Fig. 2. System dynamics model: Stock and flow diagram.



Fig. 3. Recycling capacity increment time impact on recycling capacity.



Fig. 4. Charging frequency impact on EOL LIB supply.

soundness and suitability of the model for the following reasons: (1) Our model resembles the behavior of the general structure of existing simulation models of the automotive recycling and repurposing market (Bobba et al., 2019; Dunn et al., 2022; Wasesa et al., 2022; Kamath et al., 2023; Chirumalla et al., 2023; Shafique et al., 2023). The research mentioned above highlights that a higher collection rate leads to higher raw material recovery and increases the availability of repurposed LIB. The results highlighted by them resemble the results explained in Sections 4.2, 4.3, 4.4, and 4.5. (2) The equations that establish the interconnection among the model variables are checked for their similarity with the existing literature and variables dimensional consistency. (3) Equations connecting model variables are based on well-defined theories, such as learning effect, capacity expansion, etc., defined in the literature.

We perform a structure-oriented behavior test on the SD model by assigning extreme values to the model variables, following the approach described by Barlas (1996). This test aims to identify and uncover any structural flaws in the model. The model is tested for the following scenarios: (1) Impact of recycling capacity increment time on recycling capacity and (2) Impact of charging frequency on EOL LIB supply. In all Scenarios, we consider the average battery capacity as 60kWh for high Cobalt scenarios explained in Table 4, collection capacity is taken as 50% (only 50% of EOL LIBs are collected), and 80% of collected EOL LIB is allocated for recycling. Fig. 3 shows the variations in recycling capacity with a change in recycling capacity increment time (RCIT). RCIT varies from 6 to 30 months (RCIT6- RCIT30), where RCIT6 means six months to increase the recycling capacity, and RCIT30 means 30 months to increase the capacity. The value of RCIT for analysis is taken as "RCIT6, RCIT12, RCIT18, RCIT24, and RCIT30". RCIT increment delays the capacity increment rate, which reduces the total recycling capacity. Fig. 3 shows that in RCIT6, recycling capacity reaches 399,600 units/month, whereas, in RCIT30, recycling capacity reaches 131,700 units/month for NMC622.

In other scenarios, the charging frequency varies between 1 and 7 cycles/week (where one cycle/week (CF1) is the minimum charging frequency, whereas seven cycles/week (CF7) is the maximum charging frequency). A lower charging frequency reduces the quantity of LIB replaced by increasing the LIB replacement time. In contrast, a high charging frequency decreases the LIB replacement time, which increases the LIB replacement by early LIB replacement. EOL LIB supply increases with an increase in LIB replacement. Hence, high replacement leads to high EOL LIB supply, as shown in Fig. 4.

# 4.2. Scenario 1: Variation in EOL LIB collection rate

Collection capacity depends on EOL LIB supply, minimum collection quantity (based on government policies for EOL LIB collection) (Wasesa et al., 2022; Kamath et al., 2023; Albertsen et al., 2021) and time taken to increase capacity. As the EOL LIB supply from EOL EV, SES, LIB replacement, and repurposed EOL LIB increases with time, the EOL LIB supply may exceed the collection capacity (depending on the minimum collection ratio), which increases the EOL LIB landfills. The government increases the quota for minimum collection ratio to reduce the landfill, leading to increases in EOL LIB collection. An increase in the collection rate and a high fraction of EOL LIB allocated to recycling increases the recovery of raw materials and the recycling profitability. High profitability increases the fraction of EOL LIB allocation to the recycling process, which increases the difference between EOL LIB supply and recycling capacity. Recycling capacity increment depends on the difference between recyclable LIB supply and recycling capacity and the time taken to increment the recycling supply. Hence, an increment in difference leads to an increment in recycling capacity, further increasing the raw material recovery.

In the given scenario, "CR10- CR90" denotes the different minimum collection ratio value. Fig. 5(a) shows the recovery of Li, Fig. 5(b) shows the recovery of Ni, and Fig. 5(c) shows the recovery of Co for different minimum collection rate values. Steep changes in the recovery of raw materials signify the time taken to review collection capacity discrepancy between collection capacity, minimum collection quantity, and time to increment collection capacity and time taken to review recycling capacity discrepancy between recyclable EOL LIB supply, recycling capacity, and time to increment recycling capacity. Li, Ni, and Co recovery is high in a high recovery scenario "CR90". In the CR90 scenario, recovery of Li reduces 16% of demand, whereas Ni and Co reduce demand by 17% of total raw materials for EV and SES. Li, Ni, and Co recovery is two times higher in the CR90 scenario compared to the CR50 scenario at the end of the simulation. In the CR10 scenario, around 2% of total raw materials for EV and SES are recovered for a given proportion of EOL LIB allocated to recycling.











(c) Co recovery

Fig. 5. Collection capacity variation impact on raw material recovery from EOL LIB.





(a) Li recovery



## (b) Ni recovery

# (c) Co recovery

Fig. 6. Impact of different variant compositions in collected EOL LIB on raw material recovery.

# 4.3. Scenario 2: Variation in LIB variant composition in collected EOL LIB

EOL LIBs stock contains different LIB variants that differ from each other based on electrodes raw material composition (Nitta et al., 2015; Pillot, 2019; Vaalma et al., 2018; Bibra et al., 2021; Liu et al., 2023). Differences in the share of LIB variants in EOL LIB (Refer to Section 3.3, Table 4), raw material composition of LIB variants (Refer to Section 3.3, Table 5), allocation of EOL LIB to recycling, availability of infrastructure for EOL LIB collection and EOL LIB handing impact the raw material recovery (Refer to Section 3.3). Variability in raw material composition in LIB electrode and different LIB variant demand influences the raw material recovery from EOL LIB as raw material recovery depends on both.

In this scenario, we analyze the impact of varying proportions of LIB variants in collected EOL LIB battery mix on raw material recovery. Figs. 6(a), 6(b), and 6(c) show variation in Li, Ni, and Co recovery with variation in LIB variants proportion in collected EOL LIB, respectively. Steep changes in the recovery of raw materials signify the increment in the difference between EOL LIB supply and collection capacity and recyclable LIB supply and recycling capacity similar to Scenario 4.3. Fig. 6(b) shows that Ni recovery is high in high Nickel scenarios as NMC811 LIB demand dominates the LIB demand. Figs. 6(c) and 6(a) show that Co and Li demand (respectively) is high in high cobalt scenarios, as NMC622 LIB dominates the overall LIB demand. Recovery of raw materials reduces the demand for total raw materials for EV and SES by 7%-10% based on differences in LIB variants composition in overall LIB demand (as defined in Table 4). Li demand is reduced by 5%-9%, Ni demand is reduced by 6%-11%, and the Co demand by 5%-10% based on the difference in LIB variants proportion in battery mix, recycling efficiency, and proportion of EOL LIB allocated to recycling. Co recovery is three times, and Ni recovery is around 1.5 times higher in high cobalt scenarios compared to LFP scenarios as LFP does not contain Ni and Co.

#### 4.4. Scenario 3: Variation in the repurposing on second use LIB

Repurposing EOL LIB creates a secondary supply, reducing the demand for new LIB for energy storage. The availability of repurposed LIB variants (depends on different scenarios defined in Table 4) relies on the quantity of EOL LIB allocated for the repurposing and repurposing infrastructure. In this scenario, EOL LIB allocated for the repurposing process varies from 10%–90%. "RR1-RR9" where RR1 denotes the 10% of the total collected EOL LIBs allotted to the repurposing process, and RR9 denotes the 90% of the total collected EOL LIBs allotted to the repurposing process. In this scenario, "RR1, RR3, RR5, RR7, and RR9" are taken into consideration. A large quantity of EOL LIB allocation to repurposing increases the availability of LIB to be repurposed for stationary storage applications, limited by the available capacity for the repurposing process.

Fig. 7 shows the variations in the quantity of NMC811 battery obtained by repurposing the EOL LIBs. Repurposed LIB quantity increases with the increase in the EOL LIB collection, the allocation of a large share of EOL LIB (RR9) to repurposing, and adequate repurposing infrastructure. In the RR9 scenario, the LIB available for second use is around three times higher than the total repurposed LIB available in the RR3 scenario. In RR9, the availability of repurposed NMC622 is 2.7 times, and NMC822 and NCA are 2.9 times higher than the respective repurposed LIB variable available for the RR3 scenario. Recyclable NMC622, NMC811, and NCA increased by 0.2M units, 0.03M units, and 0.027M units due to the EOL repurposed LIB supply in 2030. An increment in recyclable depends on the battery mix of EOL LIB.



Fig. 7. Variation in the quantity of LIB available for secondary application.

#### 4.5. Scenario 4: Variation in the repurposing on raw material recovery

Repurposing of EOL LIB reduces the demand for new LIB for energy storage by supplying repurposed LIB similar to Scenario 3 (Section 4.4). More repurposed EOL LIB leads to a large reduction of new LIB demand for energy storage. However, repurposing delayed the recovery of raw materials from EOL LIB as it extended the LIB life. Hence, the higher allocation of EOL LIB to repurposing reduces the quantity of recovered raw materials as more repurposing reduces the LIB allocated for recycling.

In this scenario, RR1 signifies that 10% of collected EOL LIB is allocated to repurposing facilities, whereas RR9 denotes 90% of collected EOL LIB is allocated to repurposing. Similar to Scenario 4.4, we take "RR1, RR3, RR5, RR7, and RR9" into consideration. Figs. 8(a), 8(b), and 8(c) show the variations in Li, Ni, and Co recovery, respectively, with the variation of allocation of EOL LIB to repurposing. Increment in allocation to EOL LIB to repurposing facilities leads to lower recovery of raw materials as shown in Figure 8(a)–8(c). RR9 leads to 3%-4%recovery of Ni, Co, and Ni of total Li, Co, and Ni demand from EV and SES, whereas RR1 leads to recovery of 10%-11% of total raw material demand. Focussing on the RR9 scenario, in 2035, about 3500 ton Li, 19260 ton Ni, and 5500 ton Co will be stocked in repurposed LIB.

# 5. Discussion

The system dynamics modeling technique is used to understand the impact of different EOL LIB handling processes on the continually increasing EOL LIB stock with different LIB variant proportions on raw material recovery. Different scenarios related to varying EOL LIB collection rates, allocation of the different LIB variants (LIB proportion in EOL LIB) to the recycling and repurposing process, EOL LIB handling process infrastructure, and delay in the development of infrastructure are assessed to analyze their impact on the raw material recovery.

#### 5.1. Comparison with other studies' findings

Our study mainly supports and extends the findings of other studies. Model results indicate that variation in collection rate impacts the collection of different LIB variants based on the battery mix of EOL LIB, which leads to variation in raw material recovery. Simulation results indicate that varying collection rates from 10% to 90% (CR10 to CR90) will reduce the demand for total raw materials Ni and Co by (1.8–17)%. In contrast, demand for Li is reduced by (1.7%–16%) for a different proportion of EOL LIB allocated to the recycling and repurposing process, comparable to results obtained by Xu et al. (2020), Dunn













Fig. 8. Repurposing impact on raw material recovery from EOL LIB.

et al. (2022). The percentage of recovered material is less compared to results obtained by Liu et al. (2023) due to differences in assumptions of the model (for EOL LIB calculation (Liu et al., 2023) considers EV, consumer electronics, uninterruptible power supply, and energy storage system). Recycling will provide the recycled material when LIB reaches EOL, whereas repurposing delays the material recovery as repurposed LIB is used for secondary application (Fig. 2). Recyclable LIB variant supply is reduced by 0.027M-0.2M (based on the high cobalt scenario defined in Table 4) when EOL LIBs are allocated to repurposing as repurposing delayed the LIB to recycling based on LIB warranty life for repurposed application. Due to repurposing (RR3 Scenario), LIB addresses to recycling in 2035 are estimated to be 1.4 times lower than those of Scenario when all the EOL LIB allocated to recycling. Reduction in recyclable quantity is slightly higher than Bobba et al. (2019) (LIB address to recycling in 2035 are estimated to be 1.25 times lower) because Bobba et al. (2019) consider a gradual increase in repurposing till 20%. In contrast, we considered 30% of EOL LIB allocated to repurposing.

Along with collection capacity and the trade-off between recycling and repurposing, the availability of different LIB variants in EOL LIB also impacts the recovery of raw materials from EOL LIB, affecting raw material demand. Variation in LIB variants demand defined in the High Nickel scenario in Table 4 leads to 41% lower recovery of Co compared to high cobalt scenarios. In contrast, material recovery of Li, Ni, and Co is low for the LFP scenario compared to other scenarios in Table 4 due to the difference in the proportion of constituent's raw materials in LIB variants contained in EOL LIB stock (Xu et al., 2020; Dunn et al., 2022; Kamath et al., 2023).

## 5.2. Theoretical contributions

This study provides a model to analyze the tradeoffs between different circular economy (CE) strategies used for raw material recovery from EOL LIBs that contain different LIB variants (variants differ in raw material compositions of electrodes). Most of the literature assumes that LIB replacement that impacts EOL LIB stock depends on the LIB warranty life and excludes the impact of charging frequency on LIB replacement (Shafique et al., 2022; Huster et al., 2024). Our analyses overcome this assumption by highlighting that LIB replacement depends on LIB warranty life and charging frequency. EOL LIBs are collected from different vehicles to adhere to government policy regarding collection rates to reduce landfills. Our model results highlight that EOL LIB collection infrastructure and battery mix in EOL LIB impact the collection of different LIB variants that will affect the raw material recovery from EOL LIB in the future based on adopting different CE strategies like recycling and repurposing. Repurposing enhances the longevity of EOL LIB by extending LIB utilization in applications like energy storage, peak shaving, back-up, frequency regulation, and renewables integration (Chen et al., 2019). It adds to the recyclable LIB supply stock when it reaches their EOL.

EOL LIB stock segregation for recycling and repurposing depends on the profitability of both processes. Many frameworks highlight that recycling profitability depends on the recyclable quantity, which depends on recycling capacity, recycling process efficiency, battery capacity, recovery of raw material, and raw materials cost. In contrast, repurposing profitability depends on the repurposed quantity that depends on repurposing capacity, battery capacity, and cost of repurposed battery per kWh. Our model results highlight that along with capacity, time delay and incremental change in capacity impact the profitability of the EOL LIB handling process, which influences the EOL LIB allocation to recycling and repurposing and, hence, raw material recovery.

## 5.3. Policy implications

Previous studies have shown that multiple stakeholders interact to ensure the proper handling of EOL LIB (Glöser-Chahoud et al., 2021; Yang et al., 2024). This study proposes an SD modeling approach that provides a conceptual framework to analyze the impact of interaction among multiple stakeholders involved with different CE strategies that enable raw material recovery. Hence, this framework will help policymakers contextualize the impact of tradeoffs between different CE strategies on raw material recovery from EOL LIB. Raw material recovery from EOL LIB depends on LIB variant proportion in EOL LIB stock, the chemical composition of LIB variant's electrodes, and collection rate (Liu et al., 2023; Shafique et al., 2023). Our model simulation results highlight that the battery mix composition of EOL LIB combined with high collection leads to high raw material recovery. Therefore, battery manufacturers' choice of LIB variants and policies supporting improvement in the collection facility infrastructure improves material recovery, reducing recycling and repurposing costs per kWh. LIB variants such as LFP yield lesser recovery, so they can be routed to repurposing and energy storage. This will put forward the need for a proper policy framework that enables separate collection and sorting of LIBs for different variants.

Apart from LIB variant mix and collection capacity infrastructure, tradeoffs between CE strategies like recycling and repurposing, the efficiency of the handling process, and recycling and repurposing infrastructure will also impact raw materials recovery. Hence, countries like India with scarce sources of strategic raw materials (like Li, Ni, and Co) and in the nascent stage of LIB manufacturing should frame policies focused on incentivizing better collection and EOL LIB handling techniques to alleviate the raw material recovery and availability of repurposed LIB for SESs to reduce the import dependency on raw material and LIB.

# 5.4. Limitations

Although the findings regarding the examined variables align with the existing literature, the variables are limited to the collection rate, consumer preferences for LIB variants, and allocation of EOL LIB for recycling and repurposing. In this study, we consider only the impact of the market share of NMC, NCA, and LFP in EOL LIB stock on raw material recovery. Battery technologies are undergoing revolutionary breakthroughs to develop LIBs like lithium-air, lithium-sulfur, sodiumion, and solid-state batteries (Zeng et al., 2022; Ren et al., 2023). Adopting these LIB chemistries in the future will impact the raw material recovery and second use of EOL LIB. Therefore, future investigation is required to explore the impact of market share on the EOL LIB handling process for future LIB chemistries.

This study employs SD modeling to understand the impact of tradeoffs between the EOL LIB handling process and raw material recovery. However, this study needs to account for the effects of changes in policy framework on EOL LIB handling infrastructure development. Additionally, the EOL LIB handling process has environmental (Richa et al., 2017; Kamath et al., 2020) and economic impact (Xiong et al., 2020; Schulz-Mönninghoff and Evans, 2023) based on the quality assessment of the EOL LIB. Changes in the regulatory framework either support or hinder the infrastructure development that will impact the environmental and economic feasibility of the EOL LIB handling process, thereby affecting the recovery of raw materials.

# 6. Conclusions

This study utilizes system dynamics modeling to understand the variability in raw material recovery based on scenarios on EOL collection rates, LIB variants, and repurposing rates. Based on the analysis, we conclude that EOL LIB collection infrastructure improvement and regulation framework on EOL LIB collection reduces the landfill by enabling better EOL LIB collection, consumer preferences for materialrich LIB variants may prove beneficial with higher recovery rates compared to others, and the EOL LIB stock exhibits an upward trend with increasing demand for LIB in EVs and energy storage, with repurposed LIB partially fulfilling the LIB demand for energy storage. Continual improvement of the EOL LIB handling process, collection infrastructure, and selection of proper CE strategies plays a crucial role in the future security of LIB raw material demand. Recycling improves the raw material supply by recovering raw material from EOL LIB, whereas repurposing enhances LIB longevity as repurposed LIB is used to meet the SES demand. The LIB variant proportion in EOL LIB and their electrods' composition also impact raw material recovery. Thus, original equipment manufacturers (OEMs) involved in the EOL LIB handling business should plan their strategies by considering the demand for material-intensive LIB.

This study considers LIB's demand for electric cars and SES. The model can be enhanced by considering EV variants like electric buses, two/three-wheelers, and electric trucks. This research considered only NMC, NCA, and LFP LIB chemistries for analysis. Therefore, this research can be extended to cover the impact of future batteries on the recovery process. Along with different LIB variants, collection facilities, recycling, and repurposing infrastructure also impact material recovery. Improvement in infrastructure needs investment and transparent government policies. This study can be extended by including the social factors and policy differences between developed and emerging economies' impact on the CE strategies used for handling EOL LIB. The recent Conference of Parties (COP27) and policies adopted by countries like the U.S. and different parts of Europe for stringent emission standards have refocused the global effort on developing sustainable technologies for clean mobility, carbon neutrality, etc. The model boundary can be extended by adding the impact of policy aspects such as government emission targets, manufacturers, and government future EV sales target commitment to establishing the circular economy for handling EOL LIBs.

# CRediT authorship contribution statement

Bhanu Pratap: Writing – review & editing, Writing – original draft, Visualization, Validation, Software, Methodology, Conceptualization. T.V. Krishna Mohan: Writing – review & editing, Writing – original draft, Software, Methodology, Conceptualization. R.K. Amit: Writing – review & editing, Writing – original draft, Supervision, Methodology, Conceptualization. Shankar Venugopal: Supervision, Conceptualization.

## Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

# Acknowledgments

We gratefully acknowledge the valuable comments from the participants of the OR Society's 64th Annual Conference and the 2022 Winter Simulation Conference. We are also deeply indebted to the anonymous editor and reviewers of the journal for their valuable comments.

# Appendix A. Supplementary data

Supplementary material related to this article can be found online at https://doi.org/10.1016/j.spc.2024.07.027.

# References

- Abbas, K.A., Bell, M.G., 1994. System dynamics applicability to transportation modeling. Transp. Res. A 28 (5), 373–390. http://dx.doi.org/10.1016/0965-8564(94) 90022-1.
- Abergel, T., Bunsen, T., Gorner, M., Leduc, P., Pal, S., 2020. Global EV Outlook 2020 Entering the Decade of Electric Drive? Technical Report, International Energy Agency.
- Alamerew, Y.A., Brissaud, D., 2020. Modelling reverse supply chain through system dynamics for realizing the transition towards the circular economy: A case study on electric vehicle batteries. J. Clean. Prod. 254, 120025. http://dx.doi.org/10. 1016/J.JCLEPRO.2020.120025.
- Albertsen, L., Richter, J.L., Peck, P., Dalhammar, C., Plepys, A., 2021. Circular business models for electric vehicle lithium-ion batteries: An analysis of current practices of vehicle manufacturers and policies in the EU. Resour. Conserv. Recy. 172, 105658. http://dx.doi.org/10.1016/J.RESCONREC.2021.105658.
- Alfaro-Algaba, M., Ramirez, F.J., 2020. Techno-economic and environmental disassembly planning of lithium-ion electric vehicle battery packs for remanufacturing. Resour. Conserv. Recy. 154, 104461. http://dx.doi.org/10.1016/J.RESCONREC. 2019.104461.
- Alsauskas, O., Connelly, E., Daou, A., Gouy, A., HUismans, M., Kim, H., Teter, J., 2023. Global EV Outlook 2023: Catching Up with Climate Ambitions. Technical Report, International Energy Agency (IEA), URL www.iea.org.
- Alšauskas, O., Connelly, E., Mathilde, H., Jenness, E., 2024. Global EV Outlook 2024 Moving towards increased affordability. Technical Report, International Energy Agency, URL www.iea.org/reports/global-ev-outlook-2024.
- Barlas, Y., 1989. Multiple tests for validation of system dynamics type of simulation models. European J. Oper. Res. 42 (1), 59–87. http://dx.doi.org/10.1016/0377-2217(89)90059-3.
- Barlas, Y., 1996. Formal aspects of model validity and validation in system dynamics. Syst. Dyn. Rev. 12 (3), 183–210. http://dx.doi.org/10.1002/(sici)1099-1727(199623)12:3<183::aid-sdr103>3.0.co;2-4.
- Berends, P.A., Romme, A.G., 2001. Cyclicality of capital-intensive industries: a system dynamics simulation study of the paper industry. Omega 29 (6), 543–552. http: //dx.doi.org/10.1016/S0305-0483(01)00043-3.
- Beuse, M., Steffen, B., Schmidt, T.S., 2020. Projecting the competition between energystorage technologies in the electricity sector. Joule 4 (10), 2162–2184. http: //dx.doi.org/10.1016/j.joule.2020.07.017.
- Bhanu, P., Krishna Mohan, T.V., Amit, R.K., Shankar, V., 2024. Factors affecting the market dynamics of lithium-ion battery for electric mobility: a system dynamics perspective. J. Simul. 18 (3), 428–445. http://dx.doi.org/10.1080/17477778.2022. 2150578.
- Bibra, E.M., Connelly, E., Dhir, S., Drtil, M., Cozzi, L., 2022. Global EV Outlook 2022 Securing Supplies for an Electric Future. Technical Report, International Energy Agency, URL www.iea.org/t&c/.
- Bibra, E.M., Gul, T., Connelly, E., Gorner, M., Teter, J., 2021. Global EV Outlook 2021 Accelerating Ambitions Despite the Pandemic. Technical Report, International Energy Agency, URL www.iea.org/t&c/.
- Bobba, S., Mathieux, F., Ardente, F., Blengini, G.A., Cusenza, M.A., Podias, A., Pfrang, A., 2018a. Life Cycle Assessment of repurposed electric vehicle batteries: an adapted method based on modelling energy flows. J. Energy Storage 19, 213–225. http://dx.doi.org/10.1016/J.EST.2018.07.008.
- Bobba, S., Mathieux, F., Blengini, G.A., 2019. How will second-use of batteries affect stocks and flows in the EU? A model for traction Li-ion batteries. Resour. Conserv. Recy. 145, 279–291. http://dx.doi.org/10.1016/J.RESCONREC.2019.02.022.
- Bobba, S., Podias, A., Persio, F.D., Messagie, M., Tecchio, P., Cusenza, M.A., Eynard, U., Mathieux, F., Pfrang, A., 2018b. Sustainability Assessment of Second Life Application of Automotive Batteries (SASLAB). Technical Report, Joint Research Centre(JRC), European Commission's science and knowledge service, p. 140. http:// dx.doi.org/10.2760/53624, JRC112543, URL https://core.ac.uk/reader/162256840.
- Chen, M., Ma, X., Chen, B., Arsenault, R., Karlson, P., Simon, N., Wang, Y., 2019. Recycling end-of-life electric vehicle lithium-ion batteries. Joule 3 (11), 2622–2646. http://dx.doi.org/10.1016/J.JOULE.2019.09.014.
- Chen, J., Zhang, W., Gong, B., Zhang, X., Li, H., 2022. Optimal policy for the recycling of electric vehicle retired power batteries. Technol. Forecast. Soc. Change 183, 121930. http://dx.doi.org/10.1016/J.TECHFORE.2022.121930.
- Chirumalla, K., Kulkov, I., Vu, F., Rahic, M., 2023. Second life use of li-ion batteries in the heavy-duty vehicle industry: Feasibilities of remanufacturing, repurposing, and reusing approaches. Sustain. Prod. Consumpt. 42, 2352–5509. http://dx.doi. org/10.1016/j.spc.2023.10.007.
- Dai, Q., Spangenberger, J., Ahmed, S., Gaines, L., Kelly, J.C., Wang, M., 2019. EverBatt: A Closed-loop Battery Recycling Cost and Environmental Impacts Model Energy Systems Division. Technical Report, p. 124, URL www.anl.gov.
- Dunn, J., Kendall, A., Slattery, M., 2022. Electric vehicle lithium-ion battery recycled content standards for the US – targets, costs, and environmental impacts. Resour. Conserv. Recy. 185, 106488. http://dx.doi.org/10.1016/J.RESCONREC.2022. 106488.
- Engel, H., Hertzke, P., Siccardo, G., 2019. Second-Life EV batteries: The Newest Value Pool in Energy Storage. Technical Report, URL https: //www.mckinsey.com/industries/automotive-and-assembly/our-insights/secondlife-ev-batteries-the-newest-value-pool-in-energy-storage.

- Feng, Y.Y., Chen, S.Q., Zhang, L.X., 2013. System dynamics modeling for urban energy consumption and CO2 emissions: A case study of Beijing, China. Ecol. Model. 252 (1), 44–52. http://dx.doi.org/10.1016/J.ECOLMODEL.2012.09.008.
- Fleischmann, J., Hanicke, M., Horetsky, E., Ibrahim, D., Jautelat, S., Linder, M., Schaufuss, P., Torscht, L., van de Rijt, A., 2023. Battery 2030: Resilient, Sustainable, and Circular. Technical Report, URL https://www.mckinsey. com/industries/automotive-and-assembly/our-insights/battery-2030-resilientsustainable-and-circular.
- Ford, A., 1999. Cycles in competitive electricity markets: A simulation study of the western United States. Energy Policy 27 (11), 637–658. http://dx.doi.org/10.1016/ S0301-4215(99)00050-6.
- Ford, A., 2018. Simulating systems with fast and slow dynamics: lessons from the electric power industry. Syst. Dyn. Rev. 34 (1–2), 222–254. http://dx.doi.org/10. 1002/SDR.1595.
- Ford, A., Flynn, H., 2005. Statistical screening of system dynamics models. Syst. Dyn. Rev. 21 (4), 273–303. http://dx.doi.org/10.1002/SDR.322.
- Gaines, L., 2014. The future of automotive lithium-ion battery recycling: Charting a sustainable course. Sustain. Mater. Technol. 1–2, 2–7. http://dx.doi.org/10.1016/ J.SUSMAT.2014.10.001.
- Georgiadis, P., Athanasiou, E., 2013. Flexible long-term capacity planning in closedloop supply chains with remanufacturing. European J. Oper. Res. 225 (1), 44–58. http://dx.doi.org/10.1016/J.EJOR.2012.09.021.
- Georgiadis, P., Besiou, M., 2008. Sustainability in electrical and electronic equipment closed-loop supply chains: A System Dynamics approach. J. Clean. Prod. 16 (15), 1665–1678. http://dx.doi.org/10.1016/J.JCLEPRO.2008.04.019.
- Georgiadis, P., Besiou, M., 2010. Environmental and economical sustainability of WEEE closed-loop supply chains with recycling: a system dynamics analysis. Int. J. Adv. Manuf. Technol. 475–493. http://dx.doi.org/10.1007/s00170-009-2362-7.
- Glöser-Chahoud, S., Huster, S., Rosenberg, S., Baazouzi, S., Kiemel, S., Singh, S., Schneider, C., Weeber, M., Miehe, R., Schultmann, F., 2021. Industrial disassembling as a key enabler of circular economy solutions for obsolete electric vehicle battery systems. Resour. Conserv. Recycl. (174), 105735. http://dx.doi.org/10. 1016/j.resconrec.2021.105735.
- Goodenough, J.B., 2012. Rechargeable batteries: Challenges old and new. J. Solid State Electrochem. 16 (6), 2019–2029. http://dx.doi.org/10.1007/s10008-012-1751-2.
- Harper, G., Sommerville, R., Kendrick, E., Driscoll, L., Slater, P., Stolkin, R., Walton, A., Christensen, P., Heidrich, O., Lambert, S., Abbott, A., Ryder, K., Gaines, L., Anderson, P., 2019. Recycling lithium-ion batteries from electric vehicles. Nature 575 (7781), 75–86. http://dx.doi.org/10.1038/s41586-019-1682-5.
- Harrison, G., Thiel, C., Jones, L., 2016. Powertrain Technology Transition Market Agent Model (PTTMAM). In: JRC Technical Reports. European Commission, http: //dx.doi.org/10.2790/719385, URL https://ec.europa.eu/jrc.
- Heymans, C., Walker, S.B., Young, S.B., Fowler, M., 2014. Economic analysis of second use electric vehicle batteries for residential energy storage and load-levelling. Energy Policy 71, 22–30. http://dx.doi.org/10.1016/J.ENPOL.2014.04.016.
- Hogler, D., Petroni, G., 2022. Old Electric-Vehicle Batteries Are Getting a Second Life. Wall Street J. URL https://www.wsj.com/articles/old-electric-vehicle-batteries-aregetting-a-second-life-11655114401.
- Huster, S., Rosenberg, S., Hufnagel, S., Rudi, A., Schultmann, F., 2024. Do consumers want reconditioned electric vehicle batteries? – A discrete choice experiment. Sustain. Prod. Consumpt. 48, 446–459. http://dx.doi.org/10.1016/J.SPC.2024.05. 027.
- Jacoby, M., 2019. It's time to get serious about recycling lithium-ion batteries.
- Jiang, S., Zhang, L., Hua, H., Liu, X., Wu, H., Yuan, Z., 2021. Assessment of end-of-life electric vehicle batteries in China: Future scenarios and economic benefits. Waste Manage. 135, 70–78. http://dx.doi.org/10.1016/J.WASMAN.2021.08.031.
- Kamath, D., Moore, S., Arsenault, R., Anctil, A., 2023. A system dynamics model for end-of-life management of electric vehicle batteries in the US: Comparing the cost, carbon, and material requirements of remanufacturing and recycling. Resour. Conserv. Recycl. 196, 107061. http://dx.doi.org/10.1016/j.resconrec.2023.107061.
- Kamath, D., Shukla, S., Arsenault, R., Kim, H.C., Anctil, A., 2020. Evaluating the cost and carbon footprint of second-life electric vehicle batteries in residential and utility-level applications. Waste Manage. 113, 497–507. http://dx.doi.org/10.1016/ J.WASMAN.2020.05.034.
- Kamran, M., Raugei, M., Hutchinson, A., 2021. A dynamic material flow analysis of lithium-ion battery metals for electric vehicles and grid storage in the UK: Assessing the impact of shared mobility and end-of-life strategies. Resour. Conserv. Recy. 167, 105412. http://dx.doi.org/10.1016/J.RESCONREC.2021.105412.
- Kastanaki, E., Giannis, A., 2023. Dynamic estimation of end-of-life electric vehicle batteries in the EU-27 considering reuse, remanufacturing and recycling options. J. Clean. Prod. 393, 136349. http://dx.doi.org/10.1016/j.jclepro.2023.136349.
- Khowaja, A., Dean, M.D., Kockelman, K.M., 2022. Quantifying the emissions impact of repurposed electric vehicle battery packs in residential settings. J. Energy Storage 47, 103628. http://dx.doi.org/10.1016/J.EST.2021.103628.
- Krishna Mohan, T.V., Pratap, B., Amit, R.K., Venugopal, S., 2021. A Road Map to Sustainable Mobility: Analyzing the Dynamics of Lithium-Ion Battery Recycling. In: 2021 IEEE Transportation Electrification Conference. ITEC-India 2021, IEEE, http://dx.doi.org/10.1109/ITEC-India53713.2021.9932513.
- Kunc, M.H., Morecroft, J.D., 2007. Competitive dynamics and gaming simulation: lessons from a fishing industry simulator. J. Oper. Res. Soc. 58 (9), 1146–1155. http://dx.doi.org/10.1057/PALGRAVE.JORS.2602246.

- Lander, L., Cleaver, T., Rajaeifar, M.A., Nguyen-Tien, V., Elliott, R.J., Heidrich, O., Kendrick, E., Edge, J.S., Offer, G., 2021. Financial viability of electric vehicle lithium-ion battery recycling. iScience 24 (7), 102787. http://dx.doi.org/10.1016/ J.ISCI.2021.102787.
- Li, L., Dababneh, F., Zhao, J., 2018. Cost-effective supply chain for electric vehicle battery remanufacturing. Appl. Energy 226, 277–286. http://dx.doi.org/10.1016/ J.APENERGY.2018.05.115.
- Li, Y., Liang, C., Ye, F., Zhao, X., 2023. Designing government subsidy schemes to promote the electric vehicle industry: A system dynamics model perspective. Transp. Res. A 167, 103558. http://dx.doi.org/10.1016/J.TRA.2022.11.018.
- Li, X., Mu, D., Du, J., Cao, J., Zhao, F., 2020. Game-based system dynamics simulation of deposit-refund scheme for electric vehicle battery recycling in China. Resour. Conserv. Recy. 157, 104788. http://dx.doi.org/10.1016/J.RESCONREC. 2020.104788.
- Liu, S., Dong, Z., Ding, C., Wang, T., Zhang, Y., 2020. Do you need cobalt ore? Estimating potential trade relations through link prediction. Resour. Policy 66, 101632. http://dx.doi.org/10.1016/j.resourpol.2020.101632.
- Liu, M., Liu, W., Liu, W., Chen, Z., Cui, Z., 2023. To what extent can recycling batteries help alleviate metal supply shortages and environmental pressures in China? Sustain. Prod. Consumpt. 36, 139–147. http://dx.doi.org/10.1016/j.spc. 2023.01.004.
- Maisel, F., Neef, C., Marscheider-Weidemann, F., Nissen, N.F., 2023. A forecast on future raw material demand and recycling potential of lithium-ion batteries in electric vehicles. Resour. Conserv. Recy. 192, 106920. http://dx.doi.org/10.1016/ J.RESCONREC.2023.106920.
- Makwarimba, C.P., Tang, M., Peng, Y., Lu, S., Zheng, L., Zhao, Z., Zhen, A.g., 2022. Assessment of recycling methods and processes for lithium-ion batteries. iScience 25 (5), 104321. http://dx.doi.org/10.1016/J.ISCI.2022.104321.
- Martinez-Laserna, E., Gandiaga, I., Sarasketa-Zabala, E., Badeda, J., Stroe, D.I., Swierczynski, M., Goikoetxea, A., 2018. Battery second life: Hype, hope or reality? A critical review of the state of the art. Renew. Sustain. Energy Rev. 93, 701–718. http://dx.doi.org/10.1016/J.RSER.2018.04.035.
- Mohan, K., Amit, R.K., 2021. Sustain. Prod. Consumpt. Modeling oligopsony market for end-of-life vehicle recycling. Sustain. Prod. Consumpt. 25, 325–346. http: //dx.doi.org/10.1016/j.spc.2020.09.001.
- Morse, I., 2021. A dead battery dilemma. Science 372 (6544), 780–783. http://dx.doi.org/10.1126/SCIENCE.372.6544.780/ASSET/0DA07CAA-1A4E-4827-887C-37576105A8DF/ASSETS/GRAPHIC/372\_780\_F4.JPEG, URL https: //www.science.org/doi/10.1126/science.372.6544.780.
- Neubauer, J., Smith, K., Wood, E., Pesaran, A., 2015. Identifying and Overcoming Critical Barriers to Widespread Second Use of PEV Batteries. Technical Report, URL www.nrel.gov/publications.
- Nieuwenhuijsen, J., Correia, G.H.d., Milakis, D., van Arem, B., van Daalen, E., 2018. Towards a quantitative method to analyze the long-term innovation diffusion of automated vehicles technology using system dynamics. Transp. Res. C 86, 300–327. http://dx.doi.org/10.1016/J.TRC.2017.11.016.
- Nitta, N., Wu, F., Lee, J.T., Yushin, G., 2015. Li-ion battery materials: Present and future. Mater. Today 18 (5), 252–264. http://dx.doi.org/10.1016/j.mattod.2014. 10.040.
- Olivetti, E.A., Ceder, G., Gaustad, G.G., Fu, X., 2017. Lithium-Ion Battery Supply Chain Considerations: Analysis of Potential Bottlenecks in Critical Metals. Joule 1 (2), 229–243. http://dx.doi.org/10.1016/j.joule.2017.08.019.
- Pillot, C., 2019. The Rechargeable Battery Market and Main Trends 2018–2030. Technical Report, 36th Annual International Battery Seminar & Exhibit. Avicenne Energy.
- Pratap, B., Krishna Mohan, T.V., Amit, R.K., Venugopal, S., 2019. Visualizing the complex demand and supply dynamics of critical battery materials - A key enabler for realizing the Electric Mobility Vision 2030. In: 2019 IEEE Transportation Electrification Conference. ITEC-India 2019, http://dx.doi.org/10.1109/ITEC-India48457. 2019.ITECIndia2019-170.
- Pratap, B., Krishna Mohan, T.V., Amit, R.K., Venugopal, S., 2023. Analyzing the impact of lithium-ion battery recovery strategy on e-mobility revolution. In: 2023 IEEE International Transportation Electrification Conference. ITEC-India, pp. 1–5. http://dx.doi.org/10.1109/ITEC-India59098.2023.10471418.
- Reinhardt, R., Christodoulou, I., Gassó-Domingo, S., Amante García, B., 2019. Towards sustainable business models for electric vehicle battery second use: A critical review. J. Environ. Manag. 245, 432–446. http://dx.doi.org/10.1016/j.jenvman. 2019.05.095.
- Ren, Z., Li, H., Yan, W., Lv, W., Zhang, G., Lv, L., Sun, L., Sun, Z., Gao, W., 2023. Comprehensive evaluation on production and recycling of lithium-ion batteries: A critical review. Renew. Sustain. Energy Rev. 185, 113585. http://dx.doi.org/10. 1016/j.rser.2023.113585.
- Richa, K., Babbitt, C.W., Gaustad, G., Wang, X., 2014. A future perspective on lithiumion battery waste flows from electric vehicles. Resour. Conserv. Recy. 83, 63–76. http://dx.doi.org/10.1016/j.resconrec.2013.11.008.
- Richa, K., Babbitt, C.W., Nenadic, N.G., Gaustad, G., 2017. Environmental tradeoffs across cascading lithium-ion battery life cycles. Int. J. Life Cycle Assess. 22 (1), 66–81. http://dx.doi.org/10.1007/S11367-015-0942-3/FIGURES/7, URL https: //link.springer.com/article/10.1007/S11367-015-0942-3.

- Rizova, M.I., Wong, T.C., Ijomah, W., 2020. A systematic review of decision-making in remanufacturing. Comput. Ind. Eng. 147, 106681. http://dx.doi.org/10.1016/J. CIE.2020.106681.
- Ryan, P., 2021. Old electric cars are a raw material of the future. The Economist, URL https://www.economist.com/science-and-technology/2021/05/13/old-electriccars-are-a-raw-material-of-the-future.
- Schmuch, R., Wagner, R., Hörpel, G., Placke, T., Winter, M., 2018. Performance and cost of materials for lithium-based rechargeable automotive batteries. Nat. Energy 3 (4), 267–278. http://dx.doi.org/10.1038/s41560-018-0107-2.
- Schulz-Mönninghoff, M., Bey, N., Nørregaard, P.U., Niero, M., 2021. Integration of energy flow modelling in life cycle assessment of electric vehicle battery repurposing: Evaluation of multi-use cases and comparison of circular business models. Resour. Conserv. Recv. 174, http://dx.doi.org/10.1016/J.RESCONREC.2021.105773.
- Schulz-Mönninghoff, M., Evans, S., 2023. Key tasks for ensuring economic viability of circular projects: Learnings from a real-world project on repurposing electric vehicle batteries. Sustain. Prod. Consumpt. 35, 559–575. http://dx.doi.org/10.1016/J.SPC. 2022.11.025.
- Senge, P.M., Sterman, J.D., 1992. Systems thinking and organizational learning: Acting locally and thinking globally in the organization of the future. European J. Oper. Res. 59 (1), 137–150. http://dx.doi.org/10.1016/0377-2217(92)90011-W.
- Shafique, M., Ateeq, M., Rafiq, M., Azam, A., Luo, X., 2023. Prospects of recycling from end-of-life of Li-ion batteries on alleviating materials demand-supply gap in new electric vehicles in Asia. Waste Manage. 171, 207–217. http://dx.doi.org/10. 1016/j.wasman.2023.08.033.
- Shafique, M., Rafiq, M., Azam, A., Luo, X., 2022. Material flow analysis for end-oflife lithium-ion batteries from battery electric vehicles in the USA and China. Resour. Conserv. Recy. 178, 106061. http://dx.doi.org/10.1016/J.RESCONREC. 2021.106061.
- Slattery, M., Dunn, J., Kendall, A., 2021. Transportation of electric vehicle lithium-ion batteries at end-of-life: A literature review. Resour. Conserv. Recy. 174, 105755. http://dx.doi.org/10.1016/J.RESCONREC.2021.105755.
- Sommerville, R., Zhu, P., Rajaeifar, M.A., Heidrich, O., Goodship, V., Kendrick, E., 2021. A qualitative assessment of lithium ion battery recycling processes. Resour. Conserv. Recy. 165, 105219. http://dx.doi.org/10.1016/J.RESCONREC.2020.105219.Sterman, J.D., 2010. Business dynamics. McGraw Hill Higher Education, p. 982.
- Sverdrup, H.U., Ragnarsdottir, K.V., 2016. A system dynamics model for platinum group metal supply, market price, depletion of extractable amounts, ore grade, recycling and stocks-in-use. Resour. Conserv. Recy. 114, 130–152. http://dx.doi.
- org/10.1016/j.resconrec.2016.07.011.
  Taylor, T.R., Ford, D.N., Ford, A., 2010. Improving model understanding using statistical screening. Syst. Dyn. Rev. 26 (1), 73–87. http://dx.doi.org/10.1002/SDR.428.
- Thies, C., Kieckhäfer, K., Spengler, T.S., 2016. Market introduction strategies for alternative powertrains in long-range passenger cars under competition. Transp. Res. D 45, 4–27. http://dx.doi.org/10.1016/j.trd.2015.05.002.
- Vaalma, C., Buchholz, D., Weil, M., Passerini, S., 2018. A cost and resourse analysis of sodium-ion battery. Nat. Rev. Mater. 3, 18013. http://dx.doi.org/10.1038/ natrevmats.2018.13.

- Vlachos, D., Georgiadis, P., Iakovou, E., 2007. A system dynamics model for dynamic capacity planning of remanufacturing in closed-loop supply chains. Comput. Oper. Res. 34 (2), 367–394. http://dx.doi.org/10.1016/J.COR.2005.03.005.
- Wang, Y., Chang, X., Chen, Z., Zhong, Y., Fan, T., 2014. Impact of subsidy policies on recycling and remanufacturing using system dynamics methodology: a case of auto parts in China. J. Clean. Prod. 74, 161–171. http://dx.doi.org/10.1016/J.JCLEPRO. 2014.03.023.
- Wasesa, M., Hidayat, T., Andariesta, D.T., Natha, M.G., Attazahri, A.K., Afrianto, M.A., Mubarok, M.Z., Zulhan, Z., Putro, U.S., 2022. Economic and environmental assessments of an integrated lithium-ion battery waste recycling supply chain: A hybrid simulation approach. J. Clean. Prod. 379, http://dx.doi.org/10.1016/J.JCLEPRO. 2022.134625.
- Weigl, D., Young, D., 2023. Impact of automated battery sorting for mineral recovery from lithium-ion battery recycling in the United States. Resour. Conserv. Recy. 192, 106936. http://dx.doi.org/10.1016/J.RESCONREC.2023.106936.
- White, C., Thompson, B., Swan, L.G., 2020. Repurposed electric vehicle battery performance in second-life electricity grid frequency regulation service. J. Energy Storage 28, http://dx.doi.org/10.1016/J.EST.2020.101278.
- Wolstenholme, E.F., Coyle, R.G., 1983. The development of system dynamics as a methodology for system description and qualitative analysis. J. Oper. Res. Soc. 34 (7), 569. http://dx.doi.org/10.2307/2581770.
- Xiong, S., Ji, J., Ma, X., 2020. Environmental and economic evaluation of remanufacturing lithium-ion batteries from electric vehicles. Waste Manage. 102, 579–586. http://dx.doi.org/10.1016/J.WASMAN.2019.11.013.
- Xu, C., Behrens, P., Gasper, P., Smith, K., Hu, M., Tukker, A., Steubing, B., 2023. Electric vehicle batteries alone could satisfy short-term grid storage demand by as early as 2030. Nature Commun. 14 (1), 1–11. http://dx.doi.org/10.1038/s41467-022-35393-0.
- Xu, C., Dai, Q., Gaines, L., Hu, M., Tukker, A., Steubing, B., 2020. Future material demand for automotive lithium-based batteries. Commun. Mater. 1 (1), 1–10. http://dx.doi.org/10.1038/s43246-020-00095-x.
- Yang, D., Wang, M., Luo, F., Liu, W., Chen, L., Li, X., 2024. Evaluating the recycling potential and economic benefits of end-of-life power batteries in China based on different scenarios. Sustain. Prod. Consumpt. 47, 145–155. http://dx.doi.org/10. 1016/J.SPC.2024.04.001.
- Zeng, A., Chen, W., Dalgas Rasmussen, K., Zhu, X., Lundhaug, M., Müller, D.B., Tan, J., Keiding, J.K., Liu, L., Dai, T., Wang, A., Liu, G., 2022. Battery technology and recycling alone will not save the electric mobility transition from future cobalt shortages. Nature Commun. 13, 1341. http://dx.doi.org/10.1038/s41467-022-29022-z.
- Zhu, J., Mathews, I., Ren, D., Li, W., Cogswell, D., Xing, B., Sedlatschek, T., Kantareddy, S.N.R., Yi, M., Gao, T., Xia, Y., Zhou, Q., Wierzbicki, T., Bazant, M.Z., 2021. End-of-life or second-life options for retired electric vehicle batteries. Cell Rep. Phys. Sci. 2 (8), 100537. http://dx.doi.org/10.1016/J.XCRP.2021.100537.
- Zubi, G., Dufo-López, R., Carvalho, M., Pasaoglu, G., 2018. The lithium-ion battery: State of the art and future perspectives. Renew. Sustain. Energy Rev. 89, 292–308. http://dx.doi.org/10.1016/j.rser.2018.03.002.